Author's response to reviews

Title: The effect of a sports chiropractic manual therapy intervention on the prevention of back pain, hamstring and lower limb injuries in semi-elite Australian Rules footballers: a randomized controlled trial

Authors:

Wayne Hoskins (waynehoskins@iinet.net.au)
Henry Pollard (hpollard@optushome.com.au)

Version: 5 Date: 31 March 2010

Author's response to reviews:

Corrected one grammatical error and a spelling error